EPIDEMIOLOGY AND HEALTH SERVICES RESEARCH

Clinical performance of the Nevisense system in cutaneous melanoma detection: an international, multicentre, prospective and blinded clinical trial on efficacy and safety*

J. Malvehy,¹ A. Hauschild,² C. Curiel-Lewandrowski,³ P. Mohr,⁴ R. Hofmann-Wellenhof,⁵ R. Motley,⁶ C. Berking,⁷ D. Grossman,⁸ J. Paoli,⁹ C. Loquai,¹⁰ J. Olah,¹¹ U. Reinhold,¹² H. Wenger,¹³ T. Dirschka,¹⁴ S. Davis,¹⁵ C. Henderson,¹⁶ H. Rabinovitz,¹⁷ J. Welzel,¹⁸ D. Schadendorf¹⁹ and U. Birgersson^{20,21}

¹Department of Dermatology, Hospital Clinic, IDIBAPS, University of Barcelona, Barcelona, Spain

²Department of Dermatology, University of Kiel, Kiel, Germany

³Department of Dermatology, Arizona Cancer Center, University of Arizona, Tucson, AZ, U.S.A.

⁴Department of Dermatology, Elbe-Klinikum Buxtehude, Buxtehude, Germany

⁵Department of Dermatology, University of Graz, Graz, Austria

⁶Department of Dermatology, University Hospital of Wales, Cardiff, U.K.

⁷Department of Dermatology, Ludwig-Maximilian University, Munich, Germany

⁸Department of Dermatology, University of Utah, Salt Lake City, UT, U.S.A.

⁹Department of Dermatology, Sahlgrenska University Hospital, Gothenburg, Sweden

¹⁰Department of Dermatology, University of Mainz, Mainz, Germany

¹¹Department of Dermatology, University of Szeged, Szeged, Hungary

¹²Department of Dermatology, Medical Center Bonn-Friedenplatz, Friedensplatz, Bonn, Germany

¹³Department of Dermatology, County Hospital Ryhov, Jönköping, Sweden

¹⁴Dermatologische Praxis, Wuppertal, Germany

¹⁵Dermatology Clinical Research Center, San Antonio, TX, U.S.A.

¹⁶Department of Dermatology, Southampton University Hospital, Southampton, U.K.

¹⁷Department of Dermatology, Skin and Cancer Associates, Plantation, FL, U.S.A.

¹⁸Department of Dermatology, General Hospital Augsburg, Augsburg, Germany

¹⁹Department of Dermatology, University of Essen, Essen, Germany

²⁰Division of Imaging and Technology, Department of Clinical Science, Intervention and Technology, Alfred Nobels Allé 10, Karolinska Institutet, Huddinge, SE 141 86, Sweden

²¹SciBase AB, Stockholm, Sweden

Summary

Correspondence

Ulrik Birgersson. E-mail: ulrik.birgersson@ki.se

Accepted for publication 8 May 2014

Funding sources This study was funded by SciBase AB.

Conflicts of interest

U.B. is an employee of SciBase AB. A.H., P.M. and U.R. serve as consultants to SciBase AB.

*Plain language summary available online

DOI 10.1111/bjd.13121

Background Even though progress has been made, the detection of melanoma still poses a challenge. In light of this situation, the Nevisense electrical impedance spectroscopy (EIS) system (SciBase AB, Stockholm, Sweden) was designed and shown to have the potential to be used as an adjunct diagnostic tool for melanoma detection.

Objectives To assess the effectiveness and safety of the Nevisense system in the distinction of benign lesions of the skin from melanoma with electrical impedance spectroscopy.

Methods This multicentre, prospective, and blinded clinical study was conducted at five American and 17 European investigational sites. All eligible skin lesions in the study were examined with the EIS-based Nevisense system, photographed, removed by excisional biopsy and subjected to histopathological evaluation. A postprocedure clinical follow-up was conducted at 7 ± 3 days from the initial measurement. A total of 1951 patients with 2416 lesions were enrolled into the study; 1943 lesions were eligible and evaluable for the primary efficacy end point, including 265 melanomas – 112 in situ and 153 invasive melanomas with a median Breslow thickness of 0.57 mm [48 basal cell carcinomas (BCCs) and seven squamous cell carcinomas (SCCs)].

 $\ensuremath{\mathbb{C}}$ 2014 The Authors. British Journal of Dermatology

British Journal of Dermatology (2014) 171, pp1099-1107

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made. Results The observed sensitivity of Nevisense was 96.6% (256 of 265 melanomas) with an exact one-sided 95% lower confidence bound estimated at 94.2% and an observed specificity of 34.4%, and an exact two-sided 95% confidence bound estimated at 32.0-36.9%. The positive and negative predictive values of Nevisense were 21.1% and 98.2%, respectively. The observed sensitivity for nonmelanoma skin cancer was 100% (55 of 48 BCCs and seven SCCs) with an exact two-sided 95% confidence bound estimated at 93.5-100.0%.

Conclusions Nevisense is an accurate and safe device to support clinicians in the detection of cutaneous melanoma.

What's already known about this topic?

- Although progress has been made in the detection of melanoma it still poses a challenge.
- Electrical impedance spectroscopy (EIS) may potentially be used as a diagnostic aid for the detection of melanoma.

What does this study add?

- In the largest international prospective study of its kind in melanoma detection, the EIS system Nevisense was shown to be both accurate and safe in the lesion cohort studied.
- In the absence of a perfect gold standard, the accuracy of a device should be compared with the consensus diagnosis from multiple experts.

Early detection of melanoma is vital for treatment outcome and survival. Treatment of early-stage melanoma is mostly curative, whereas thicker melanomas are associated with a poor 5-year survival rate due to increased metastatic potential.^{1,2} In most instances, physicians feel fairly confident when distinguishing nonsuspicious from suspicious lesions by relying on unaided eye examination, dermoscopy assessment and patient history. However, cutaneous melanomas can be misdiagnosed as benign, and a significant proportion of benign lesions are unnecessarily excised. The sensitivity of the clinical diagnosis of cutaneous melanoma with unaided eye examination is around 60% and can improve significantly with the use of dermoscopy.³⁻³⁷ Although progress has been made in the detection of melanoma it still poses a challenge, especially in its earlier stages. Therefore, a number of technologies utilizing either visual or nonvisual techniques, such as total body photography,^{38,39} confocal microscopy,^{40,41} Raman spectroscopy,^{42,43} multispectral imaging,⁴⁴ automated dermoscopy image analysis,⁴⁵ genomic detection of melanoma by stratum corneum stripping⁴⁶ and electrical impedance spectroscopy (EIS),47,48 have been developed to support physicians in detecting melanomas at an earlier stage. In a previous study, the EIS-based Nevisense system (SciBase AB, Stockholm, Sweden) was shown to have the potential to be used as an adjunct diagnostic tool, although it was concluded that more clinical data were necessary to ensure the safety and effectiveness of the system.48

The aim of this clinical investigation was to assess the safety and effectiveness of the Nevisense system, which has been designed to aid in the discrimination between benign lesions and primary cutaneous melanoma. In this article, results are presented and discussed in the context of clinical utility.

Materials and methods

Ethical conduct

The guidelines of the revised Declaration of Helsinki, the Guidelines of Good Clinical Practice (ISO-14155), and the demands of national and data protection laws and other applicable regulatory requirements were followed. The clinical trial registration number is NCT01077050 (www.ClinicalTrials.gov).

Study design and data acquisition

Recruitment into this blinded multicentre prospective study was conducted at five American and 17 European investigational sites (Sweden, Germany, Austria, Hungary, U.K. and Spain). Potential study participants were screened according to the inclusion and exclusion criteria. Subsequent to written informed consent, patients were asked about their medical history and a clinical evaluation was performed. A photograph and dermoscopic image of each included lesion was taken before and after Nevisense measurements to document evaluation according to the protocol. In accordance with standard clinical practice, eligible and evaluable lesions were excised and subjected to the investigational site's histopathology evaluation and managed accordingly.

A further histopathological evaluation was completed by a panel of three experienced histopathologists who evaluated each lesion independently and were blinded from the investigational site's original histopathology diagnosis. In the case of agreement among the experts, the diagnosis was considered as the study's histopathological gold standard (HGS). If there was significant disagreement among the pathology reviewers on whether the lesion represented a malignancy, the respective slides were submitted to two additional experts whose diagnosis was then chosen as the HGS if they reached agreement. In case of disagreement by the two additional reviewers, the corresponding lesion was excluded from the efficacy analysis.

A postprocedure follow-up either by a telephone call or at a participant's visit to the investigational site was conducted at 7 \pm 3 days after the Nevisense evaluation, at which time the patient was evaluated for any adverse events.

Inclusion and exclusion criteria

Patients with skin lesions selected for total excision to rule out melanoma were asked to participate in the study. To minimize selection bias, all lesions destined for excision were eligible for inclusion in the study. To ensure a broad spectrum of excised lesions, dermatologists were encouraged to enroll a mix of lesions with an even distribution of low-, medium-and high-risk lesions. The exclusion criteria were derived from previous studies conducted with the investigational device and are listed in Table 1.^{47,48}

Table 1 Exclusion criteria

Men or women of any ethnic group aged < 18 years
Patient not willing or able to read, understand and sign the
study-specific informed consent form
Metastases or recurrent lesions
Lesion $< 2 \text{ mm or} > 20 \text{ mm in diameter}$
Lesion located on acral skin, e.g. sole or palm
Lesion located on areas of scars, crusts, psoriasis, eczema or
similar skin conditions
Lesion on hair-covered areas, e.g. scalp, beards, moustaches
or whiskers
Lesion located on genitalia
Lesion located in an area that has been previously biopsied or
subjected to any kind of surgical intervention or trauma
Lesion located on mucosal surfaces
Lesion with foreign matter, e.g. tattoo or splinter
Lesion and/or reference located on acute sunburn
Skin surface not measurable, e.g. lesion on a stalk
Skin surface not accessible, e.g. inside ears, under nails
Skin not intact (measurement area), e.g. bleeding or with
clinical noticeable ulceration

Review of images

The photographs and dermoscopic images were taken with a Sony DSC-W290 (Sony, Tokyo, Japan) and a hand-held dermoscope (DermLite II PRO HR[®], 3Gen; DermLite, San Juan Capistrano, CA, U.S.A.). The images were reviewed by three dermatologists with 2–5 years of experience in dermoscopy assessment. The option to reach out to additional experienced dermoscopists in difficult cases was allowed. Lesions were classified according to the clinical ABCD rule,^{49,50} the dermoscopic ABCD rule,⁵¹ the seven-point checklist⁵² and the overall suspicion of malignancy classified by the visual classification board from 0 (benign) to 10 (malignant). This was conducted to ascertain a standardized clinical and dermoscopic classification of the degree of suspicion of malignancy of each study lesion.

Blinding

The case report forms, the Nevisense measurements and the histopathological reports were kept blinded from the sponsor by a contract research organization until classification of all lesions in the pivotal study had been made.

The investigators were blinded to the entire diagnostic information of the device to ensure that the device could not bias the clinical judgement nor affect the clinical management of the patient in any way.

Electrical impedance spectroscopy measurements

Electrical impedance was measured with the Nevisense system equipped with a spring-loaded probe and a disposable electrode having an active area of approximately 5 \times 5 mm². Prior to measurement, the skin was moistened for 30 s with physiological saline solution, after which a reference measurement of healthy skin close to the lesion was performed. The procedure was then repeated on the lesion under study. The system measures the overall electrical resistance and reactance at 35 different frequencies logarithmically distributed between 1.0 kHz and 2.5 MHz at four depth settings with a total of 10 permutations. The applied voltage and resulting current is limited to 150 mV and 75 $\mu A,$ respectively, and is not sensed by the patient. Measurements take approximately 8 s, and within seconds the system computes both a score (0-10) and a dichotomous output (EIS negative/positive) at a fixed cut-off. The fixed threshold is set at 4, i.e. scores < 4 are EIS negative and scores of \geq 4 are EIS positive. The dichotomous output was used in the study to demonstrate the sensitivity and specificity end points.

Study objective and end point

The objective of this clinical investigation was to determine the safety and effectiveness of the Nevisense device, which has been designed to help distinguish between cutaneous melanoma and benign lesions of the skin, using EIS relative to the HGS. This study had two coprimary analyses, aiming to demonstrate the accuracy of the Nevisense device: (i) a one-sided exact 95% confidence bound of the sensitivity in detecting cutaneous melanoma of > 90% (sensitivity \geq 0.90 to detect melanoma); (ii) nonrandom result at the given sensitivity, i.e. sensitivity + specificity > 1.0.

The safety of the Nevisense analysis was measured by the occurrence and incidence of all adverse events reported for study participants throughout their participation in the study. The primary safety end point was achieved if no serious adverse events related to the device had occurred.

Additional analysis

Clinical histopathological gold standard

To estimate the sensitivity and specificity of the clinical HGS used for treatment the investigational site's histopathological diagnoses were compared with the study HGS. The analysis was conducted on eligible and evaluable lesions.

Unaided lesion evaluation and dermoscopy assessment

The observed sensitivity and specificity of the visual reference standard was calculated using the cut-offs prespecified from literature and the outcome of the different visual classification algorithms.

Results

Study participants and skin lesions

A total of 1951 participants with 2416 lesions were recruited between March 2010 and November 2011. The demographic characteristics of the study population are presented in Table 2. The median age of the patients was 48 years (range 18–91). Women comprised 51.9% of the study group. Most (97.1%) patients were white and the majority were either of Fitzpatrick skin type 2 or 3. No significant differences in the demographic characteristics between the enrolled and the eligible lesions were observed.

Table 3 presents the distribution of reasons for exclusion from the effectiveness analysis. Out of the 2416 registered lesions a total of 473 were excluded, mainly owing to investigator oversight or the inability to render a final histopathological diagnosis. Approximately one-quarter of the excluded lesions were device-related (inadequate reference measurement quality or general device failures).

Performance of Nevisense

The dichotomous outcome of the Nevisense system was compared with the HGS. Of the 1943 eligible and evaluable lesions (Table 4), 265 (13.2%) were cutaneous melanoma, 55 (2.8%) were nonmelanoma skin cancer (NMSC), including
 Table 2 Demographic characteristics of the enrolled and eligible participants

	Patients enrolled	Patients with eligible lesions
Characteristics	(n = 1950)	(n = 1611)
Sex		
Male	929 (47.6)	765 (47.5)
Female	1013 (51.9)	846 (52.5)
Missing	8 (0.4)	0 (0)
Age (years),	48 (18-91)	48 (18-91)
median (range)		
Race and ethnicity		
Asian	5 (0.3)	5 (0.3)
White	1893 (97.1)	1571 (97.5)
Black or African	2 (0.1)	2 (0.1)
American		
Hispanic or Latino	29 (1.5)	25 (1.6)
Other	12 (0.6)	8 (0.5)
Missing	9 (0.5)	0 (0)
Fitzpatrick skin type		
1. Always burns	136 (7)	117 (7.3)
easily; never tans		
2. Always burns	945 (48.5)	783 (48.6)
easily; tans minimally		
3. Burns moderately;	635 (32.6)	526 (32.7)
tans gradually	. ,	
4. Burns minimally;	192 (9.8)	158 (9.8)
always tans well		~ /
5. Rarely burns;	29 (1.5)	23 (1.4)
tans profusely	~ /	× /
6. Never burns;	1 (0.1)	1 (0.1)
deeply pigmented		
Missing	12 (0.6)	3 (0.2)

Data presented as number (percentage) of patients, except for age. For one subject, the signed informed consent form could not be located at the site and the data were thus deleted.

basal cell carcinomas (BCCs) and squamous cell carcinomas (SCCs), of which Nevisense correctly identified 256 melanomas and all 55 NMSCs, yielding an observed sensitivity of $96{\cdot}6\%$ and $100{\cdot}0\%$, respectively. A total of 157 naevi with severe dysplasia were included, of which Nevisense gave a positive reading for malignancy in 132 cases. Seven out of eight actinic keratoses gave a positive reading. One Merkel cell carcinoma was included, which was correctly identified as malignant. Out of the remaining 1457 lesions, 501 were diagnosed as negative, yielding an observed specificity of 34.4%. No significant difference in the presented sensitivity and specificity was encountered, when the possible dependency in outcome between the lesions of the same patient was accounted for through a generalized linear mixed model. The positive predictive value (PPV) of Nevisense was 21.1% and the negative predictive value (NPV) was 98.2%. The Nevisense score was compared with lesion severity and, as can be discerned from Figure 1, a clear step function is evident for the score outcome with increasing lesion severity.

Table 3 Reasons for exclusion of	of lesions from the analysi	S
----------------------------------	-----------------------------	---

	No. of lesions		
Reason for exclusion	n	%	Source
Lesions included	2416		
Signed informed consent form missing	1	< 0.1	Investigator: 11.0%
Withdrawal	17	0.7	
Not eligible (i.e.	61	2.5	
Major protocol violation	29	1.2	
Measurement not acquired	60	2.5	
Coverage ^a	98	$4 \cdot 1$	
Ineligible histopathology (preparation quality)	8	0.3	Pathology: 4·1%
Missing histopathology ^b	39	1.6	
Inaccurate mapping of histopathology ^c	7	0.3	
No consensus ^d	44	1.8	
Poor reference quality ^e	95	3.9	Device-related: 4.5%
Device failure	14	0.6	
Eligible lesions	1943		

^aLess than 75% of the lesion was covered with measurements, e.g. a 10 \times 10-mm² lesion only measured once was excluded. ^bNo histology slides were/could be provided by the site. ^cProvided histology slides were not mapped accurately to the lesion measured. ^dThe consensus board could not agree on a final diagnosis. ^eInability to obtain a reference measurement with adequate quality after four consecutive attempts.

Nevisense false negatives

Of the nine melanomas classified by Nevisense as false negative, seven had sufficient image quality to render an outcome for the ABCD dermoscopy rule and seven-point checklist by the reviewing dermatologists. The overall visual malignancy grading scored two of this subset of seven melanomas as positive with a score of 4, and five as negative with a score of 0-2.

Three of the nine FN lesions were identified on patients aged < 30 years. The median diameter of these nine lesions was 4 mm (range 2–8), of which five had a diameter < 6 mm. Seven were in situ and two were early invasive melanomas (T1a) with Breslow thicknesses of 0.4 mm and 0.6 mm, respectively. The lesions were located on the following anatomical locations: lower extremities (n = 4), upper extremities (n = 1), upper back (n = 2), buttocks (n = 1) and facial area (n = 1).

Performance of the unaided examination and dermoscopy assessment

To determine the diagnostic uncertainty of the study lesions, a postexcisional performance study of unaided examination and dermoscopy was completed. Of the 1943 eligible and evaluable lesions, 1701 (238 melanomas) had sufficient image

Table 4 Observed sensitivity (Sens) and specificity (Spec) for Nevisense combined with lower (LCB) and upper 95% confidence bounds (UCB), and true and false positives/negatives in the efficacy analysis population differentiated by histopathological lesion type and melanoma thickness

Туре	Sens	TP^{a}	FN ^a	Total	$\mathrm{LCB}^{\mathrm{b}}$	UCB^b
Melanoma ^c	96.6	256	9	265	$93{\cdot}7^{\rm d}$	98.4
pTis	93.8	105	7	112	87.6	97.5
pT1a	97.9	92	2	94	92.5	99.7
pT1b	100	19	0	19	82.4	100
pT2a	100	24	0	24	85.8	100
pT2b	100	11	0	11	71.5	100
pT3a	100	1	0	1	2.5	100
pT3b	100	3	0	3	29.2	100
pT4a	100	1	0	1	2.5	100
pT4b	NA	0	0	0	NA	NA
Severe dysplasia ^e	84.1	132	25	157	77.4	89.4
Туре	Spec	TN ^a	FP ^a	Total	LCB ^b	UCB ^b
Mild/moderate dysplasia	36.1	357	631	988	33.1	39.2
Moderate dysplasia	24.1	80	252	332	19.6	29.1
Mild dysplasia	41.3	212	301	513	37.0	45.7
Dysplastic Lentigo	40.0	2	3	5	5.3	85.3
Structural disorder only	73.3	11	4	15	44.9	92.2
Undecided ^f	42.3	52	71	123	33.4	51.5
Melanocytic naevus	36.7	131	226	357	31.7	41.9
Blue naevus	24.0	6	19	25	9.4	45.1
Compound naevus	34.0	33	64	97	24.7	44.3
Dermal naevus	38.5	37	59	96	28.8	49.0
Halo naevus	42.9	3	4	7	9.9	81.6
Junction naevus	85.7	6	1	7	42.1	99.6
Lentigo	55.0	11	9	20	31.5	76.9
Other	33.3	9	18	27	16.5	54.0
Reed naevus	31.3	5	11	16	11.0	58.7
Spitz naevus	0	0	5	5	0	52.2
Undecided	36.8	21	36	57	24.5	50.7
Other	11.6	13	99	112	6.3	19.0
Lichenoid keratosis	0	0	4	4	0	60.2
Seborrhoeic keratosis	7.8	4	47	51	2.2	18.9
Other	15.8	9	48	57	7.5	27.9
Overall specificity	34.4	501	956	1457	32.0	36.9
Туре	Sens	TP ^a	FN ^a	Total	LCB ^b	UCB ^b
NMSC	100	55	0	55	93.5	100
BCC	100	48	0	48	92.6	100
SCC	100	7	0	7	59.0	100
SCC in situ	100	6	0	6	54.1	100
SCC invasive	100	1	0	1	2.5	100
Actinic keratosis	87.5	7	1	8	47.4	99.7
Merkel cell carinoma	100	1	0	1	2.5	100

TP, true positive; TN, true negative; FP, false positive; FN, false negative; NA, not applicable; NMSC, nonmelanoma skin cancer; BCC, basal cell carcinoma; SCC, squamous cell carcinoma. ^aTP correctly identified as positive, FN incorrectly identified as negative, TN correctly identified as negative and FP incorrectly identified as positive by the investigational device. ^bExact LCB and UCB calculated using the Clopper–Pearson method. ^cAmerican Joint Committee on Cancer staging system for melanoma sixth edition was used with *a* h∞ adoption of the seventh edition during the course of the study when pronounced mitosis was present.^{2,58} ^dExact Clopper–Pearson and mid-P one-sided 95% LCB estimated at 94·2% and 94·4%, respectively. ^cSevere cytological atypia or architectural disorder where diagnosed as severe dysplasia. ^fNo majority board agreement on degree of dysplasia.

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists



Fig 1. The study's histopathological gold standard plotted against the Nevisense score outcome. EIS, electrical impedance spectroscopy.

quality to enable a classification according to the ABCD rule, ABCD dermoscopy rule and the seven-point checklist.

The result of the dermoscopic and investigational site's histopathological evaluations in conjunction with the Nevisense are presented in Table 5, and grouped according to the study's HGS.

The observed sensitivity with the ABCD dermoscopy rules for melanoma detection with a cut-off of 4.75 and 5.45 was 54.2% and 47.1%, with an observed specificity of 90.1% and 94%, respectively.

The observed sensitivity for melanoma detection with the original and weighted seven-point checklists was 49.2% and 60.9% with an observed specificity of 94.2% and 89.2%, respectively.

The observed sensitivity for the reviewing dermatologist, when considering the combined features for malignancy (overall visual board malignancy grading), was 70.6%, with an observed specificity of 81.4%.

Performance of the investigational site's histopathologists

The observed sensitivity and specificity for melanoma of the investigational site's histopathologists was 84.9% (225/265) and 98.1% (1429/1457), respectively. The observed sensitivity increased with melanoma thickness, from 73.2% for Tis to 100% for T2b–T4.

Safety

A total of 36 adverse events (AEs) were observed in 28 patients (1.5%), out of which only three AEs that occurred on three patients (0.2%) were defined as definitely related to the device. All AEs related to the device were of mild severity. No serious AEs, serious adverse device effects or unanticipated adverse device effects were observed throughout the entire study.

Discussion

Melanoma detection often poses a challenge in equivocal lesions or in patients with many atypical naevi. Therefore, a component of the clinical work-up and interpretation incorporates not only lesion-specific information, but also patient-derived melanoma risk factors, which includes a comparative analysis of all lesions present on a patient.^{53–56} Even if the clinical decision is based on a collective interpretation of all the presently available clinical risk factors, melanomas can still be misdiagnosed as benign lesions.³⁷ As early detection of melanoma is vital for treatment outcome and survival,^{1,2}

Table 5 Observed sensitivity and specificity for the dermoscopic, investigational site's histopathology evaluations, as well as the Nevisense resultgrouped according to the study's histopathological gold standard (HGS). The results are derived from the cohort of eligible and evaluable lesionsthat had sufficient image quality to render a dermoscopic evaluation

Туре	Seven-point	Seven-point weighted	ABCD dermoscopy < 5.45	ABCD dermoscopy < 4.75	Malignancy grading ^a	Investigational site's histopathology	Nevisense
Melanoma sensitivity	49.2	60.9	47.1	54.2	70.6	84.5	97.1
nTis	28.7	43.6	28.7	37.6	55.4	73.3	94.1
pTIa pTIa	57.1	65.5	51.2	57.1	75.0	89.3	98.8
pT1b	76.5	76.5	76.5	82.4	88.7	100.0	100.0
pT10 pT2a	63.6	86.4	72.7	72.7	90.9	95.5	100.0
pT2b	88.9	100.0	88.9	88.9	100.0	100.0	100.0
pT3a	100.0	100.0	100.0	100.0	100.0	100.0	100.0
pT3b	100.0	100.0	66.7	100.0	100.0	100.0	100.0
pT4a	100.0	100.0	0	100.0	100.0	100.0	100.0
Severe dysplasia sensitivity ^b	12.1	24.8	12.8	20.8	38.3	NA	83.9
Overall specificity	94-2	89.2	94-0	90.1	81.4	98.0	35.8

All values given as a percentage. NA, not applicable. ^aOverall malignancy grading as determined by the visual classification board with a fixed cut-off at 4. ^bSevere cytologic atypia or architectural disorder where diagnosed as severe dysplasia.

© 2014 The Authors. British Journal of Dermatology published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists

additional objective information that could assist in the early detection of melanoma could significantly reduce the morbidity related to the unnecessary removal of benign lesions and has the potential to improve mortality through early diagnosis.

In this study, a postexcision evaluation of the isolated performance of dermoscopy was conducted according to three dermoscopic classification algorithms in which the evaluator was presented with the clinical and dermoscopic images of all excised lesions from a patient. The results show that a considerable amount of the included melanomas were equivocal lesions with insufficient dermoscopic criteria to be classified as malignant.

There are several plausible explanations for the relatively low sensitivity $(47 \cdot 1 - 70 \cdot 6\%)$ observed in the postexcisional dermoscopic evaluation with limited clinical data. Firstly, the evaluation was conducted on a cohort of $85 \cdot 0\%$ in situ and early invasive melanomas. In addition, the evaluations were completed by dermatologists with 2–5 years of experience in dermoscopy, with additional support in difficult cases. Most importantly, the evaluators did not have the added benefit of comparative analysis of all lesions present on the patients, i.e. they could only base their judgement on the excised lesions.

While the outcome of the dermoscopy assessment might have been different if the lesions had been evaluated by more experienced dermoscopists, the results of the study likely reflect the dermoscopic acumen of an average dermatologist. It has to be stressed that almost all lesions were removed owing to some clinical concern for melanoma, as it would not be ethical to excise lesions deemed benign clinically, except for functional or cosmetic reasons. As mostly preselected equivocal lesions destined for excision were included in the study, the dermoscopic diagnostic performance is not a true reflection of the performance in normal clinical practice. However, it gives an insight into the isolated performance of dermoscopy on equivocal lesions submitted for biopsy, which is the intended use of Nevisense.

Nevisense was able to achieve a high sensitivity (96.6%) in a melanoma cohort consisting mostly of in situ and early invasive melanoma without fully compromising the specificity (34.4%). The observed sensitivity of the device increased with Breslow thickness, and no invasive melanoma at stages T1b– T4 were missed by the device. Cases where the investigational device gave a false negative reading occurred mostly in small lesions with few or no dermoscopic features, and with low cellularity.

Two early-stage invasive melanomas were inaccurately classified as negative, but in both cases compliance with the measurement procedure could not be fully verified. In fact, for one lesion the verification data were missing, and the other lesion had not been fully covered with measurements. As the system only detects changes that occur directly underneath the electrode, it is important that the lesions are measured completely and/or that the most suspicious malignant part of the lesion is measured to ensure as accurate a reading as possible. A review of the 98 lesions excluded owing to coverage issues included a total of 22 melanomas, of which the device still accurately classified 20 as positive. These are good results, as for some cases only 25.0% of the surface had been measured with the device, suggesting that even though coverage is important, it is not vital in most cases.

The overall observed specificity was 34.4%. Approximately one-third of the equivocal lesions submitted for biopsy in the study were accurately identified as negative by the device and thus would not have needed a biopsy. Furthermore, the high observed NPV of 98.2% – equal to that of the observed NPV of histology in the study – ensures that few melanomas will be inaccurately left untreated when given a low score by the device.

There are two additional results that are important to highlight. Firstly, a high proportion of seborrhoeic keratoses were inaccurately classified as positive. The high ratio of positive readings was anticipated as this has been observed in previous studies, mainly owing to the typically high degree of structural changes compared with normal skin. However, the system is intended to be used by dermatologists or clinicians trained in the diagnosis of skin cancer who will almost always recognize seborrhoeic keratoses clinically and would therefore seldom apply Nevisense to these lesions. Secondly, the observed sensitivity of 100.0% in NMSC is extremely valuable as these malignancies should not be left untreated. Few other noninvasive technologies are able to identify accurately NMSC, as well as melanoma.

The observed sensitivity and specificity presented do not take into account the clinical information regarding the full extent of the patient's history, as well as the comparative analysis with other lesions, which has been shown to be critical in the clinical decision of whether to excise a lesion or not. However, in reality, clinicians often end up performing single lesion examinations owing to factors such as time constraints; as such, the isolated performance of the dermoscopy evaluation would reflect their diagnostic accuracy.⁵⁷ As can be discerned from Table 5, a large number of melanomas included in the study were equivocal lesions and hard to differentiate from nonmalignant equivocal lesions.

The Nevisense system is intended for use on cutaneous lesions with one or more clinical or historical characteristics of melanoma. The system is designed to be used when a clinician chooses to obtain additional information when considering excision and is not meant to be used to confirm a clinical diagnosis of melanoma. It should be used by physicians trained in the clinical diagnosis of skin cancer to ensure the system result is one element of the overall clinical assessment. Not only should the negative or positive EIS outcome be incorporated into the assessment, but also the EIS score, as it is coupled with the stage and severity of a lesion.

In summary, Nevisense has been shown to be an accurate and safe device that should be used in conjunction with the clinical risk assessment for patients with suspicion of melanoma in the intended use population. A negative or positive EIS reading in combination with the score outcome should be used as guidance for whether a lesion should be excised or not.

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists

Acknowledgment

We would like to thank Ragnar Jonell from the Department of Dermatology, Läkarhuset Gothenburg, Sweden, and Shasa Hu from the Department of Dermatology and Cutaneous Surgery, University of Miami, Miami, FL, U.S.A., for their clinical contribution. We would also like to thank Stig Ollmar from the Division of Imaging and Technology, Department of Clinical Science, Intervention and Technology, Karolinska, Sweden, for his valuable comments on the manuscript. SciBase AB, Stockholm, Sweden, is gratefully acknowledged for supporting this study financially.

References

- 1 Tsao H, Atkins M, Sober A et al. Management of cutaneous melanoma. N Engl J Med 2004; 351:998–1012.
- 2 Balch C, Gershenwald J, Soong S-J et al. Final version of 2009 AJCC melanoma staging and classification. J Clin Oncol 2009; 36:6199–206.
- 3 Soyer HP, Smolle J, Leitinger G et al. Diagnostic reliability of dermoscopic criteria for detecting malignant melanoma. *Dermatology* 1995; 190:25–30.
- 4 Binder M, Kittler H, Seeber A et al. Epiluminescence microscopybased classification of pigmented skin lesions using computerized image analysis and an artificial neural network. Melanoma Res 1998; 8:261–6.
- 5 Lorentzen H, Weismann K, Petersen CS et al. Clinical and dermatoscopic diagnosis of malignant melanoma: assessed by expert and non-expert groups. *Acta Derm Venereol* 1999; **79**:301–4.
- 6 Collas H, Delbarre M, De Preville PA et al. Dermatologists of the Postgraduate Association of Haute-Normandie. Evaluation of the diagnosis of pigmented tumors of the skin and factors leading to a decision to excise. Ann Dermatol Venereol 1999; 126:494–500.
- 7 Seidenari S, Pellicani G, Gianetti A. Digital videomicroscopy and image analysis with automatic classification for detection of thin melanomas. Melanoma Res 1999; 9:163–71.
- 8 Andreassi L, Perotti R, Rubegni P et al. Digital dermoscopy analysis for the differentiation of atypical nevi and early melanoma. Arch Dermatol 1999; 135:1459–65.
- 9 Bataille V, Sasieni P, Curley RK et al. Melanoma yield, number of biopsies and missed melanomas in a British teaching hospital pigmented lesion clinic: a 9-year retrospective study. Br J Dermatol 1999; 140:243–8.
- 10 Benelli C, Roscetti E, Dal Pozzo V. Reproducibility of a dermoscopic method (7FFM) for the diagnosis of malignant melanoma. Eur J Dermatol 2000; 10:110–14.
- 11 Stanganelli I, Serafini M, Bucchi LA. A cancer-registry-assisted evaluation of the accuracy of digital epiluminescence microscopy associated with clinical examination of pigmented skin lesions. Dermatology 2000; 200:11–16.
- 12 Ganster H, Pinz A, Röhrer R et al. Automated melanoma recognition. IEEE Trans Med Imaging 2001; 3:233–9.
- 13 Bafounta ML, Beauchet A, Aegerter P. Is dermoscopy (epiluminescence microscopy) useful for diagnosis of melanoma? Arch Dermatol 2001; 137:1343–50.
- 14 Megahed M, Schön M, Selimovic D et al. Reliability of diagnosis of melanoma in situ. Lancet 2002; 359:1921–2.
- 15 Offidani A, Simonetti O, Bernardini O et al. General practitioners' accuracy in diagnosing skin cancers. Dermatology 2002; 205:127– 30.

- 16 Kittler H, Pehamberger H, Wolff K et al. Diagnostic accuracy of dermoscopy. Lancet Oncol 2002; 3:159–65.
- 17 English D, Burton R, del Mar C et al. Evaluation of aid to diagnosis of pigmented skin lesions in general practice: controlled trial randomised by practice. BMJ 2003; **327**:1–6.
- 18 Argenziano G, Soyer HP, Chimenti S et al. Dermoscopy of pigmented skin lesions: results of a consensus meeting via the Internet. J Am Acad Dermatol 2003; 48:679–93.
- 19 Carli P, Nardini P, Crocetti E et al. Frequency and characteristics of melanomas missed at a pigmented lesion clinic: a registry-based study. Melanoma Res 2004; 14:403–7.
- 20 Youl P, Baade P, Janda M et al. Diagnosing skin cancer in primary care: how do mainstream general practitioners compare with primary care skin cancer clinic doctors? Med J Aust 2007; 187:215– 20.
- 21 Carli P, De Giorgi V, Crocetti E et al. Improvement of malignant/ benign ratio in excised melanocytic lesions in the 'dermoscopy era': a retrospective study 1997–2001. Br J Dermatol 2004; 150:687–92.
- 22 English DR, Del Mar C, Burton RC. Factors influencing the number needed to excise: excision rates of pigmented lesions by general practitioners. *Med J Aust* 2004; 180:16–19.
- 23 Robinson JK, Nickoloff BJ. Digital epiluminescence microscopy monitoring of high-risk patients. Arch Dermatol 2004; 140:49– 56.
- 24 Banky JP, Kelly JW, English DR et al. Incidence of new and changed nevi and melanomas detected using baseline images and dermoscopy in patients at high risk for melanoma. Arch Dermatol 2005; 141:998–1006.
- 25 Soares T, Laman S, Yiannias J et al. Factors leading to the biopsy of 1547 pigmented lesions at Mayo Clinic, Scottsdale, Arizona, in 2005. Int J Dermatol 2009; 48:1053–6.
- 26 Chen SC, Pennie ML, Kolm P et al. Diagnosing and managing cutaneous pigmented lesions: primary care physicians versus dermatologists. J Gen Intern Med 2006; 21:678–82.
- 27 Byrnes P, Ackermann E, Williams I et al. Management of skin cancer in Australia. A comparison of general practice and skin cancer clinics. Aust Fam Physician 2007; 36:1073–5.
- 28 Nathansohn N, Orenstein A, Trau H et al. Pigmented lesions clinic for early detection of melanoma: preliminary results. Isr Med Assoc J 2007; 9:708–12.
- 29 Fuller SR, Bowen GM, Tanner B et al. Digital dermoscopic monitoring of atypical nevi in patients at risk for melanoma. Dermatol Surg 2007; 33:1198–206.
- 30 Vestergaard ME, Macaskill P, Holt PE et al. Dermoscopy compared with naked eye examination for the diagnosis of primary melanoma: a meta-analysis of studies performed in a clinical setting. Br J Dermatol 2008; 159:669–76.
- 31 Heal CF, Raasch BA, Buettner PG et al. Accuracy of clinical diagnosis of skin lesions. Br J Dermatol 2008; 159:661–8.
- 32 Menzies S, Emery J, Staples M et al. Impact of dermoscopy and short-term sequential digital dermoscopy imaging for the management of pigmented lesions in primary care: a sequential intervention trial. Br J Dermatol 2009; 161:1270–7.
- 33 Warshaw EM, Lederle FA, Grill JP et al. Accuracy of teledermatology for pigmented neoplasms. J Am Acad Dermatol 2009; 61:753-65.
- 34 Rosendahl C, Tschandl P, Cameron A et al. Diagnostic accuracy of dermatoscopy for melanocytic and nonmelanocytic pigmented lesions. J Am Acad Dermatol 2011; 64:1068–73.
- 35 Salerni G, Carrera C, Lovatto L et al. Characterization of 1152 lesions excised over 10 years using total-body photography and digital dermatoscopy in the surveillance of patients at high risk for melanoma. J Am Acad Dermatol 2012; **67**:836–45.

- 36 Salerni G, Terán T, Puig S et al. Meta-analysis of digital dermoscopy follow-up of melanocytic skin lesions: a study on behalf of the International Dermoscopy Society. J Eur Acad Dermatol Venereol 2013; 27:805–14.
- 37 Guitera P, Menzies S. State of the art of diagnostic technology for early-stage melanoma. Expert Rev Anticancer Ther 2011; 11:715– 23.
- 38 Terushkin V, Oliviera S, Marghoob A et al. Use of and beliefs about total body photography and dermatoscopy among US dermatology training programs: an update. J Am Acad Dermatol 2010; 62:794– 803.
- 39 Salerni G, Carrera C, Lovatto L et al. Benefits of total body photography and digital dermatoscopy ("two-step method of digital follow-up") in the early diagnosis of melanoma in patients at high risk for melanoma. J Am Acad Dermatol 2012; 67:17-27.
- 40 Curiel-Lewandrowski C, Williams CM, Swindells KJ et al. Use of in vivo confocal microscopy in malignant melanoma: an aid in diagnosis, and assessment of surgical and non-surgical therapeutic approaches. Arch Dermatol 2004; **140**:1127–32.
- 41 Guitera P, Menzies SW, Longo C et al. In vivo confocal microscopy for diagnosis of melanoma and basal cell carcinoma using a twostep method: analysis of 710 consecutive clinically equivocal cases. J Invest Dermatol 2012; **132**:2386–94.
- 42 Lui H, Zhao J, McLean D et al. Real-time Raman spectroscopy for in vivo skin cancer diagnosis. Cancer Res 2012; 72:2491–500.
- 43 Boden I, Nyström J, Lundskog B et al. Non-invasive identification of melanoma with near-infrared and skin impedance spectroscopy. Skin Res Technol 2013; **19**:473–8.
- 44 Moheit G, Cognetta A, Ferris L et al. The performance of MelaFind. Arch Dermatol 2011; 147:188–94.
- 45 Menzies SW, Bischof L, Talbot H et al. The performance of Solar-Scan: an automated dermoscopy image analysis instrument for the diagnosis of primary melanoma. Arch Dermatol 2005; 141:1388–96.
- 46 Wachsman W, Morhenn V, Palmer T et al. Noninvasive genomic detection of melanoma. Br J Dermatol 2011; 164:797–806.

- 47 Åberg P, Birgersson U, Elsner P et al. Electrical impedance spectroscopy and the diagnostic accuracy for malignant melanoma. Exp Dermatol 2011; 20:648–52.
- 48 Mohr P, Birgersson U, Berking C et al. Electrical impedance spectroscopy as a potential adjunct diagnostic tool for cutaneous melanoma. Skin Research Technol 2013; 19:75–83.
- 49 Friedman RJ, Rigel DS, Kopf AW. Early detection of malignant melanoma: the role of physician examination and self-examination of the skin. *CA* Cancer J Clin 1985; **35**:130–51.
- 50 Abbasi N, Shaw H, Rigel D et al. Early diagnosis of cutaneous melanoma revisiting the ABCD criteria. JAMA 2004; **292**:2771–6.
- 51 Nachbar F, Stolz W, Merkle T et al. The ABCD rule of dermatoscopy: high prospective value in the diagnosis of doubtful melanocytic skin lesions. J Am Acad Dermatol 1994; 30:551–9.
- 52 Argenziano G, Fabbrocini G, Carli P et al. Epiluminescence microscopy for the diagnosis of doubtful melanocytic skin lesions: comparison of the ABCD rule of dermatoscopy and a new 7-point checklist based on pattern analysis. Arch Dermatol 1998; 134:1563–70.
- 53 Bataille V. Risk factors for melanoma development. Expert Rev Dermatol 2009; 4:533–9.
- 54 Grob JJ, Bonerandi JJ. The 'ugly duckling' sign: identification of the common characteristics of nevi in an individual as a basis for melanoma screening. Arch Dermatol 1998; 134:103–4.
- 55 Gachon J, Beaulieu P, Sei JF et al. First prospective study of the recognition process of melanoma in dermatological practice. Arch Dermatol 2005; 141:434–8.
- 56 Hofmann-Wellenhof R, Blum A, Wolf IH et al. Dermoscopic classification of atypical melanocytic nevi (Clark nevi). Arch Dermatol 2001; 137:1575–80.
- 57 Aldridge R, Naysmith L, Ooi E et al. The importance of a full clinical examination: assessment of index lesions referred to a skin cancer clinic without a total body skin examination would miss one in three melanoma. *Acta Derm Venereol* 2013; **93**:689–92.
- 58 Balch C, Buzaid A, Soong SJ et al. Final version of the American Joint Committee on Cancer staging system for cutaneous melanoma. J Clin Oncol 2002; 19:3635–48.